ORIGINAL ARTICLE VASCULAR SECTION

Insight from an Italian Delphi Consensus on EVAR feasibility outside the instruction for use: the SAFE EVAR Study

Pasqualino SIRIGNANO ^{1, 2} *, Gabriele PIFFARETTI ^{3, 4}, Silvia CERUTI ⁵, Massimiliano ORSO ⁶, Mario PICOZZI ⁵, Giovanna RICCI ⁷, Ascanio SIRIGNANO ⁷, Maurizio TAURINO ^{1, 8}, SAFE EVAR Collaborators [‡]

¹Vascular and Endovascular Surgery Unit, Sant'Andrea Hospital, Rome, Italy; ²Department of General and Specialistic Surgery, Sapienza University, Rome, Italy; ³Vascular Surgery, ASST Settelaghi University Teaching Hospital, Varese, Italy; ⁴Department of Medicine and Surgery, University of Insubria, Varese, Italy; ⁵Research Center for Clinical Ethics, Department of Biotechnologies and Life Sciences, University of Insubria, Varese, Italy; ⁶Istituto Zooprofilattico Sperimentale dell'Umbria e delle Marche "Togo Rosati", Perugia, Italy; ⁷School of Civil Law, University of Camerino, Camerino, Macerata, Italy; ⁸Department of Molecular and Clinical Medicine, Sapienza University, Rome, Italy

*Members are listed at the end of the paper

*Corresponding author: Pasqualino Sirignano, Vascular and Endovascular Surgery Unit, Sant'Andrea Hospital, Via di Grottarossa 1035/1039, 00189 Roma, Italy. E-mail: pasqualino.sirignano@uniroma1.it

This is an open access article distributed under the terms of the Creative Commons CC BY-NC license which allows users to distribute, remix, adapt and build upon the manuscript, as long as this is not done for commercial purposes, the user gives appropriate credits to the original author(s) and the source (with a link to the formal publication through the relevant DOI), provides a link to the license and indicates if changes were made. Full details on the CC BY-NC 4.0 are available at https://creativecommons.org/licenses/by-nc/4.0/.

ABSTRACT

BACKGROUND: The SAfety and FEasibility of standard EVAR outside the instruction for use (SAFE-EVAR) Study was designed to define the attitude of Italian vascular surgeons towards the use of standard endovascular repair (EVAR) for infrarenal abdominal aortic aneurysm (AAA) outside the instruction for use (IFU) through a Delphi consensus endorsed by the Italian Society of Vascular and Endovascular Surgery (Società Italiana di Chirurgia Vascolare ed Endovascolare – SICVE).

METHODS: A questionnaire consisting of 26 statements was developed, validated by an 18-member Advisory Board, and then sent to 600 Italian vascular surgeons. The Delphi process was structured in three subsequent rounds which took place between April and June 2023. In the first two rounds, respondents could indicate one of the following five degrees of agreement: 1) strongly agree; 2) partially agree; 3) neither agree nor disagree; 4) partially disagree; 5) strongly disagree; while in the third round only three different choices were proposed: 1) agree; 2) neither agree nor disagree; 3) disagree. We considered the consensus reached when \geq 70% of respondents agreed on one of the options. After the conclusion of each round, a report describing the percentage distribution of the answers was sent to all the participants.

RESULTS: Two-hundred-forty-four (40.6%) Italian Vascular Surgeons agreed to participate the first round of the Delphi Consensus; the second and the third rounds of the Delphi collected 230 responders (94.3% of the first-round responders). Four statements (15.4%) reached a consensus in the first rounds. Among the 22 remaining statements, one more consensus (3.8%) was achieved in the second round. Finally, seven more statements (26.9%) reached a consensus in the simplified last round. Globally, a consensus was reached for almost half of the proposed statements (46.1%).

CONCLUSIONS: The relatively low consensus rate obtained in this Delphi seems to confirm the discrepancy between Guideline recommendations and daily clinical practice. The data collected could represent the source for a possible guidelines' revision and the proposal of specific Good Practice Points in all those aspects with only little evidence available.

(*Cite this article as:* Sirignano P, Piffaretti G, Ceruti S, Orso M, Picozzi M, Ricci G, *et al.*; SAFE EVAR Collaborators. Insight from an Italian Delphi Consensus on EVAR feasibility outside the instruction for use: the SAFE EVAR Study. J Cardiovasc Surg 2024;65:273-9. DOI: 10.23736/S0021-9509.23.12906-5)

KEY WORDS: Abdominal aortic aneurysm; Endovascular aneurysm repair; Informed consent; Shared decision making; Consensus; Delphi technique.

Endovascular aneurysm repair (EVAR) has become more than an established alternative to open repair (OR) for the treatment of infrarenal abdominal aortic aneurysm (AAA), as this approach accounts for more than 75% of AAA elective repairs annually performed.¹ Although, compared to OR, EVAR is certainly associated with lower 30-day mortality and morbidity, and faster discharge,² some concerns remain about its durability, and need for reinterventions.¹⁻³

Based on preclinical engineering assessments and clinical study results, particular anatomical characteristics, specifically aortic neck diameter, length, angle, and shape, are recommended to guide patient selection for EVAR.⁴ Indeed, unfavorable anatomy, documented in 40–60% of treated AAAs, seems to be directly related to a negative procedural outcome.⁵⁻⁷

Nevertheless, in "real-world" clinical practice up to 44% of EVAR cases are performed using stent grafts outside their instruction for use (IFU) due to the presence of a "hostile" aortic neck anatomy, with acceptable short- and mid-term outcomes.⁸⁻¹⁰

Nevertheless, because EVAR durability is related to the maintenance of a seal between the stent graft and the aortic neck, some authors suggested that challenging neck could affect long-term outcomes, increasing the risk of type Ia endoleak, reintervention, and aneurysm-related mortality rates¹¹⁻¹⁵ a result of these long-term results, current guide-lines suggest limiting or even refraining from adopting EVAR in patients with challenging aortic necks.¹⁶⁻¹⁹

This 'inconsistency' between current guidelines and clinical practice raises not only technical and clinical uncertainties — especially in terms of long-term outcomes — but also ethical issues. Indeed, even when standard EVAR is technically feasible, or when patients are not eligible for OS, the ethical implications of performing an elective procedure outside the IFU cannot be ignored.

In these circumstances, it is therefore particularly important — in choosing and recommending the most appropriate procedure — to consider the patient's best interest. To this end, it is essential to include him/her in the decisionmaking process, considering his/her perspectives, values and preferences, beyond the theoretical advantages of each proposed procedure and the guidelines' indications.²⁰⁻²²

Although this type of assessment should always be considered an essential part of good clinical practice, in the case of such patients it takes on an even more important role, precisely because of the uncertainty produced by the gap between clinical practice and guidelines.

However, current evidence shows that shared decision-

making is still in its infancy in clinical practice and especially in surgery,²³ even though it has been found to confer several benefits beyond maximizing patients' autonomy. Patients' direct involvement through the proper management of the informed consent process, in fact, improves patient compliance, patient outcomes, and can help to reduce costs that can be associated with both undesirable treatment and subsequent litigation.²⁴

Based on these assumptions, we tried to assess whether a consensus exists among Italian vascular surgeons regarding the use of EVAR outside IFUs. Secondly, we investigated their attitudes and beliefs in relation to some ethical aspects, particularly those involving the management of the informed consent process in these circumstances.

Since consensus group methodologies, such as the Delphi method, are widely used to synthesize expert opinions in a systematic way when evidence is lacking, or questions are not manageable with experimental and epidemiological methods,²⁵ a panel of expert SICVE (Italian Society for Vascular and Endovascular Surgery) members was asked to express their level of agreement with technical, ethical and medico-legal statements on EVAR procedures performed outside the IFU.

Materials and methods

The SAfety and FEasibility of standard EVAR outside the instruction for use (SAFE-EVAR) Study aimed to define the attitude of Italian vascular surgeons towards the use of EVAR outside the instruction for use (IFU) through a Delphi consensus endorsed by the Italian Society of Vascular and Endovascular Surgery (Società Italiana di Chirurgia Vascolare ed Endovascolare – SICVE).

Questionnaire and Survey Procedure

A questionnaire consisting of 26 statements was developed and then validated by an 18-member Advisory Board. Furthermore, participants were asked to indicate their age range (<30, 31-40, 41-50, 51-60, >60 years old), experience in EVAR procedures (<100, 101-200, 201-300, >300 overall procedures performed), the type of hospital in which their activity was mainly performed (university hospital; public hospital; private hospital endorsed by the Italian National Health System; private hospital).

The questionnaire was sent *via* Google Forms to 600 Italian vascular surgeons who are members of SICVE. The Delphi process was structured in three subsequent rounds which took place between April and June 2023. In the first round, participants were sent the questionnaire contain-

ing all 26 statements and, for each statement, respondents could indicate one of the following five degrees of agreement: 1) strongly agree; 2) partially agree; 3) neither agree nor disagree; 4) partially disagree; 5) strongly disagree. We considered a consensus reached when \geq 70% of the interviewed agreed on one of the five answer options. After the conclusion of the first round, a report describing the percentage distribution of the answers was sent to the participants. In the second round, collaborators were sent the questionnaire containing only the statements for which no consensus was reached in the first round. Again, after its conclusion, respondents were sent a report describing the results. Finally, in the third round, the participants were sent the questionnaire containing only the statements for which consensus had not been reached in the first two rounds and, to converge consensus, the response options were limited to three: 1) agree; 2) neither agree nor disagree; 3) disagree. Again, the consensus threshold was set at \geq 70% for each response option. A summary of the results of the three rounds was finally sent to respondents.

Statistical analysis

Descriptive statistics have been reported. Categorical data was reported as counts and percentages. Survey responses collected by Google Forms were converted and analyzed using Microsoft Excel spreadsheets (Microsoft Corporation, Redmond, WA, USA).

Results

Two-hundred-forty-four (40.6%) Italian Vascular Surgeons agreed to join the Delphi Consensus; the main characteristics of the participants to the first round are shown in Table I. The second and the third rounds of the Delphi collected 230 replies, 94.3% of the first-round responders.

Twenty-six statements (S) were proposed as reported in methods; S from 1 to 17 were preeminently technical, focusing on Vascular Surgeons' personal consideration on safety and feasibility of out-IFU EVAR performed in different clinical and anatomical setting, while S from 18 to 26 were focused on informed consent process and others ethical considerations (Supplementary Digital Material 1: Supplementary Table I).

Four statements (15.4%) reached a consensus in the first round (S8 The proximal aortic neck is an important criterion for evaluating the feasibility of an EVAR procedure outside the IFU; S21 Given the inconsistency between what is stated in the guidelines and the results of clinical practice, performing an EVAR procedure outside the

TABLE I.—Demographic characteristics of respondents to the first round.

Demographics	N. (%)
Overall number of respondents	244
Gender, male	190 (77.9%)
Age	
<30 years	21 (8.6%)
31-40 years	83 (34.0%)
41-50 years	68 (27.9%)
51-60 years	43 (17.6%)
>60 years	29 (11.9%)
Experience in EVAR procedures	
<100	62 (25.4%)
101-200	54 (22.1%)
201-300	36 (14.8%)
>300	92 (37.7%)
Type of hospital	
Academic	117 (48.0%)
Public	109 (44.2%)
Private accredited by the Italian National Health System	18 (7.4%)
Private hospital	0 (0.0%)

IFU requires the physician to give the patient all available information regarding the procedure and its risks/benefits; S23 An EVAR procedure outside the IFU can only be considered after the physician has presented the patient with the best treatment option; S25 An EVAR procedure outside the IFU can only be considered after the physician has presented the patient with all available intervention (and non-intervention) options). Among the 22 remaining statements, one (3.8%) more consensus was reached in the second round (S24 An EVAR procedure outside the IFU can only be considered after the physician has presented the patient with the intervention options that he or she considers to be the safest and most effective). Finally, seven more statements (26.9%) reached a consensus in the simplified last Delphi consensus round: S2 Performing an EVAR outside the IFU affects mid/long-term outcome; S3 Performing an EVAR outside the IFU results in a worse outcome in terms of AAA Related Reintervention; S12 An EVAR procedure can be performed in the presence of a proximal aortic neck <15 mm in length; S14 An EVAR procedure can be performed in the presence of a proximal aortic neck <5 mm in length; S18 Proposing to a patient an EVAR procedure outside the IFU makes the informed consent process more complex (e.g. the need to spend more time informing the patient and/or providing him/her with more details regarding risks and benefits); S20 Given the inconsistency between what is stated in the guidelines and the results of clinical practice, performing an EVAR procedure outside the IFU requires the physician to give the patient the information about the procedure relevant to the patient in order to decide; S26 Even if the physician thinks that performing an EVAR outside the IFU is the most appropriate procedure for a particular patient, he or she must directly and actively involve the patient in the clinical decision-making process). Overall, a consensus was reached for 12 of the proposed statements (46.1%): S 2,3,8,12,14,1 8,20,21,23,24,25,26 (Supplementary Table I).

Discussion

Despite current guidelines suggest limiting or even refraining adopting EVAR out of IFU,¹⁶⁻¹⁹ in "real world" practice such kind of procedures are routinely performed with acceptable results.^{26, 27} Consequently, the aim of this Delphi consensus was to evaluate how Italian vascular surgeons technically and ethically face this inconsistency between "real world practice" and "best practice", and to collect an expert opinion to be used for guidelines' revision whenever needed.

According to Delphi results, Italian vascular surgeons agree that performing a standard EVAR procedure outside IFU may affect the mid/long term outcomes (S2, 71% agreement at round III), especially in term of AAA related reinterventions occurrence (S3, 74% agreement at round III). However, no consensus was reached regarding early outcomes, nor for AAA-related mortality (S1 and S4). Even more interestingly, half of the respondents for each round disagree with refraining from performing EVAR outside of IFU in the elective setting (S5), especially in those patients deemed unsuitable for open repair (S6) and in the elderly (S7). In other words, despite Italian guidelines strongly recommend that endoprosthesis choice should comply with the IFU,¹⁹ Italian vascular surgeons consider favorably to routinely perform out IFU EVAR procedures (at least in selected patients).

From an anatomical point of view, the role of proximal aortic neck characteristics (length, diameter, and angulation), aortic bifurcation, distal landing zone, and access vessels were evaluated.

Not surprisingly, aortic neck was immediately recognized as an important element in addressing EVAR feasibility outside IFU (S8, 79% agreement at round I), while aortic bifurcation (S9, 58% agreement at round III), iliac landing zone (S10, 63% agreement at round III), and adequate access vessels (S11, 61% agreement at round III) were non recognized as important criteria for EVAR feasibility.

More in details, Italian vascular surgeon agreed that a standard EVAR procedure could safely be performed in

case of a proximal aortic neck shorter than 15 mm (S12, 92% agreement at round III). No consensus was reached for length <10 mm (S13, 55% agreement at round III). Finally, a consensus was gained about refrain to perform standard EVAR in case of proximal neck <5 mm (S14, 70% at round III). Speculatively, these results could be interpreted as a clear indication to perform EVAR in patients with a proximal neck between 14 and 10 mm in length, a relative contraindication in case of length between 9 and 5 mm, and a clear contraindication to perform standard EVAR in necks shorter than 5mm. Collected opinions could be considered in a future guidelines revision to perform specific technical recommendation, at least as good practice point (GPP, according to guideline methodology).^{28, 29}

Besides proximal neck length, according to previous reported statements, no consensus was reached regarding severely angulated neck (S15), wide diameter (S16), and narrowed aortic bifurcation (S17). Theoretically, Delphi responders considered all those parameters by itself not a technical limit for standard EVAR, suggesting it could be safely and effectively performed and consequently proposed to the patients.

As for the ethical issues raised by performing an EVAR procedure outside the IFU (S18-S26), a clarification must first be made. In contrast to the first set of statements (S1-S17), these statements were not intended to establish a consensus among experts, but to collect data related to the knowledge and attitudes of Italian vascular surgeons in relation to specific ethical aspects, namely the involvement of the patient in the decision-making process and the management of the informed consent.

According to our findings, Italian vascular surgeons agree that proposing to a patient an EVAR procedure outside the IFU makes the informed consent process more complex (S18, 70% agreement at round III). However, the percentage of surgeons who answered "neither agree nor disagree" to this statement remains quite high (21% at round III). This result could be interpreted as an indication of a certain amount of uncertainty among respondents regarding how to manage the informed consent process, especially in particularly complex cases such as those investigated.

In terms of the information to be provided to the patient if an EVAR procedure is proposed outside the IFU (S19-S21), no agreement was reached among the respondents about the first proposed option, according to which the physician should only provide the patient with information that he or she believes the subject can understand (S19). In contrast, respondents agreed on both the second and third options, according to which the physician should provide information about the procedure relevant to the patient to decide (S20, 82% agreement in round III), but also all available information about the procedure and its risks/benefits (S21, 79% agreement in round I).

No agreement was therefore reached regarding the adoption (or non-adoption) of an approach traditionally perceived as paternalistic (S19), according to which it would be the physician's responsibility to decide what information should be given to the patient based on his/her assessment of the subject's ability to understand.

Conversely, respondents immediately agreed on the more comprehensive but, at the same time, less personalized option (S21), as it does not consider how the overabundance of information may confuse the patient and hinder, rather than help, him/her in the decision. Moreover, the idea that all the information about a procedure must be disclosed may reflect the physicians' overly cautious approach, which in some ways recalls forms of defensive medicine.

It is noteworthy that agreement was more difficult to achieve (III round) on the statement that truly values the doctor-patient relationship (S20), which is the one that considers it essential for the physician to consider in managing the informed consent process needs, values, and expectations relevant to the patient.

This difficulty also emerges from the answers given to the next statement, on the perceived importance of prior assessment of patient-relevant clinical and nonclinical preferences and outcomes (S22). No agreement was reached among respondents. In this case, the role played by those who remained uncertain (22% in Round III) may have been even more crucial, as the responses may reflect a lack of consideration of the relevance of patient preferences, but also the physicians' difficulty in evaluating these aspects.

Respondents agreed that the patient should be presented with both what the physician considers to be the best treatment option (S23, 70% agreement in Round I), the options that the physician considers to be the safest and most effective (S24, 70% agreement in Round II), and finally, all available intervention (and non-intervention) options (S25, 79% agreement in Round I). As these three statements were not presented alternatively, these results might suggest that the respondents perceive as important the presentation of all available options (including the nonintervention option) to the patient, but that they consider as equally important that the patient knows which options are the safest and most effective in the specific case, as well as the physician's opinion of the treatment option he or she considers preferable.

Finally, accordingly to last statement, Italian vascular surgeons believe that direct patient involvement in clinical decision making should be considered a requirement in all circumstances, regardless of whether the physician believes-from a technical and clinical perspective-that performing an EVAR outside of the IFU is the most appropriate procedure (S26, 75% agreement at Round III). However, even in this case, it is important to note that agreement on the matter was reached only by administering the statements in a simplified form, reinforcing the idea that there is still some uncertainty on the issue.

To summarize, we argue that the answers provided reflect a certain disposition among physicians to recognize the importance of direct patient involvement in the clinical decision-making process, but not a full awareness of the relevance of assessing preferences and even non-clinical aspects relevant to the patient.

Conclusions

The relatively low consensus rate obtained in this Delphi seems to confirm the discrepancy between Guideline recommendations and daily clinical practice. The data collected could represent the source for a possible guidelines' revision and the proposal of specific Good Practice Points in all those aspects with only little evidence available.

Likewise, although the data collected on the ethical aspects of performing EVAR procedures outside the IFU show, in general, a good predisposition of Italian vascular surgeons to involve patients in the clinical decisionmaking process, it is undeniable that areas of uncertainty and lack of knowledge about these issues remain. A more thorough training of physicians in relation to these aspects should therefore also be ensured, specifically regarding the proper management of the patient's informed consent process. In fact, proper information of the patient and his or her direct involvement in the decision-making process, besides being unavoidable ethical and legal requirements, are functional in defining the best strategy to be adopted for each specific individual, as his or her preferences, values and real needs are valued.

References

1. Wang S, Hicks CW, Malas MB. Neck diameter and inner curve seal zone predict endograft-related complications in highly angulated necks after endovascular aneurysm repair using the Aorfix endograft. J Vasc Surg 2018;67:760–9.

2. Kansagra K, Kang J, Taon MC, Ganguli S, Gandhi R, Vatakencherry G, et al. Advanced endografting techniques: snorkels, chimneys, peri-scopes, fenestrations, and branched endografts. Cardiovasc Diagn Ther 2018;8(Suppl 1):S175-83.

3. Kouvelos GN, Oikonomou K, Antoniou GA, Verhoeven EL, Katsargyris A. A Systematic Review of Proximal Neck Dilatation After Endovascular Repair for Abdominal Aortic Aneurysm. J Endovasc Ther 2017;24:59-67

4. Igari K, Kudo T, Toyofuku T, Jibiki M, Inoue Y. Outcomes following endovascular abdominal aortic aneurysm repair both within and outside of the instructions for use. Ann Thorac Cardiovasc Surg 2014;20:61-6.

5. Giménez-Gaibar A, González-Cañas E, Solanich-Valldaura T, Herranz-Pinilla C, Rioja-Artal S, Ferraz-Huguet E. Could Preoperative Neck Anatomy Influence Follow-up of EVAR? Ann Vasc Surg 2017;43:127–33.

6. Caradu C, Bérard X, Midy D, Ducasse E. Influence of Anatomic Angulations in Chimney and Fenestrated Endovascular Aneurysm Repair. Ann Vasc Surg 2017;43:104–14.

7. Speziale F, Sirignano P, Setacci F, Menna D, Capoccia L, Mansour W, et al. Immediate and two-year outcomes after EVAR in "on-label" and "off-label" neck anatomies using different commercially available de-vices, analysis of the experience of two Italian vascular centers. Ann Vasc Surg 2014;28:1892-900

8. Muhs BE, Jordan W, Ouriel K, Rajaee S, de Vries JP. Matched cohort comparison of endovascular abdominal aortic aneurysm repair with and without EndoAnchors. J Vasc Surg 2018;67:1699-707.

9. Troisi N, Torsello G. Commentary: new-generation devices and adjunctive procedures are the key elements to expanding the indications for endovascular aneurysm repair. J Endovasc Ther 2015;22:179-81

10. AbuRahma AF, DerDerian T, AbuRahma ZT, Hass SM, Yacoub M, Dean LS, et al. Comparative study of clinical outcome of endovascular aortic aneurysms repair in large diameter aortic necks (>31 mm) versus smaller necks. J Vasc Surg 2018;68:1345-1353.e1.

11. Schuurmann RC, van Noort K, Overeem SP, Ouriel K, Jordan WD Jr, Muhs BE, et al. Aortic Curvature Is a Predictor of Late Type Ia Endoleak and Migration After Endovascular Aneurysm Repair. J Endovasc Ther 2017;24:411-7.

12. Antoniou GA, Georgiadis GS, Antoniou SA, Kuhan G, Murray D. A meta-analysis of outcomes of endovascular abdominal aortic aneurysm repair in patients with hostile and friendly neck anatomy. J Vasc Surg 2013;57:527-38.

13. Brownrigg JR, de Bruin JL, Rossi L, Karthikesalingam A, Patterson B, Holt PJ, et al. Endovascular aneurysm sealing for infrarenal abdominal aortic aneurysms: 30-day outcomes of 105 patients in a single centre. Eur J Vasc Endovasc Surg 2015;50:157-64.

14. Schoretsanitis N, Argyriou C, Georgiadis GS, Lazaridis MK, Georgakarakos E. Hostile neck in abdominal aortic aneurysms: does it still exist? Vasc Endovascular Surg 2016;50:208-10.

15. Oliveira NF, Gonçalves FB, Ultee K, Pinto JP, Josee van Rijn M, Raa ST, et al. Patients with large neck diameter have a higher risk of type IA endoleaks and aneurysm rupture after standard endovascular aneurysm repair. J Vasc Surg 2019;69:783-91.

16. Chaikof EL, Dalman RL, Eskandari MK, Jackson BM, Lee WA, Mansour MA, *et al.* The Society for Vascular Surgery practice guidelines on the care of patients with an abdominal aortic aneurysm. J Vasc Surg 2018;67:2-77.e2.

17. Wanhainen A, Verzini F, Van Herzeele I, Allaire E, Bown M, Cohnert T, et al.; Esvs Guidelines Committee. Editor's Choice - European Society for Vascular Surgery (ESVS) 2019 Clinical Practice Guidelines on the Management of Abdominal Aorto-iliac Artery Aneurysms. Eur J Vasc Endovasc Surg 2019;57:8-93.

18. The National Institute for Health and Care Excellence (NICE). Abdominal Aortic Aneurysm: Diagnosis and Management [Nice Guideline 156]; 2020 [Internet]. Available from: https://www.nice.org.uk/guidance/ ng156 [cited 2024, Jan 24].

19. Pratesi C, Esposito D, Apostolou D, Attisani L, Bellosta R, Benedetto F, et al.; Italian Guidelines for Vascular Surgery Collaborators - AAA Group. Guidelines on the management of abdominal aortic aneurysms: updates from the Italian Society of Vascular and Endovascular Surgery (ŜICVE). J Cardiovasc Surg 2022;63:328-52

20. Berman L, Curry L, Gusberg R, Dardik A, Fraenkel L. Informed consent for abdominal aortic aneurysm repair: the patient's perspective. J Vasc Surg 2008;48:296-302

21. Bulder RM, Hamming JF, van Schaik J, Lindeman JH. Towards Patient Centred Outcomes for Elective Abdominal Aortic Aneurysm Repair: A Scoping Review of Quality of Life Scales. Eur J Vasc Endovasc Surg 2021:62:630-41.

22. Benzel E. Paternalism versus Shared Decision Making. World Neurosurg 2023;174:x.

23. Machin M, Van Herzeele I, Ubbink D, Powell JT. Shared Decision Making and the Management of Intact Abdominal Aortic Aneurysm: A Scoping Review of the Literature. Eur J Vasc Endovasc Surg 2023;65:839-49

24. Shay LA, Lafata JE. Where is the evidence? A systematic review of shared decision making and patient outcomes. Med Decis Making 2015;35:114-31

25. Hsu C, Sandford BA. The Delphi technique: making sense of consensus. Pract Assess, Res Eval 2007;12:1-8.

26. Bicknell CD. Standard EVAR Can Be Used in Most Challenging Necks: for the Motion. Eur J Vasc Endovasc Surg 2021;62:678-9.

27. Sirignano P, Ceruti S, Aloisi F, Sirignano A, Picozzi M, Taurino M. Is Evar Feasible in Challenging Aortic Neck Anatomies? A Technical Review and Ethical Discussion. J Clin Med 2022;11:4460.

28. Scottish Intercollegiate Guidelines Network. A Guideline Developer's Handbook; 2019 [Internet]. Available from: http://www.sign.ac.uk [cited 2024, Jan 24].

29. Centro Nazionale per l'Eccellenza Clinica, la Qualità e la Sicurezza delle Cure. Manuale metodologico per la produzione di linee guida di pratica clinica, versione 1.3.2; 2019 [Internet]. Available from: https:// siot.it/wp-content/uploads/2020/07/MM_v1.3.2_apr_2019.pdf [cited 2024, Jan 24].

Authors' contributions

Group author members

Group author members Giancarlo ACCARINO; Federico ACCROCCA; Giuseppe ALBA; Antonino ALBERTI; Vittorio ALBERTI; Sara ALLEVI; Francesco ALOISI; Bruno AMATO; Alessio AMICO; Francesco ANDREOLI; Domenico ANGILETTA; Antonio ANTICO; Lorenzo ANTICO; Michele ANTONELLO; Domenico BACCELLIERI; Giovanni BADALAMENTI; Gennaro BAFILE; Claudio BALDI; Chiara BARILLÀ; David BARILLÀ; Stefano BARTOLI; Giusi BASILE; Giuseppe BATTAGLIA; Cesare BATTOCCHIO; Ailin BELLONI; Raffello BELLOSTA; Domenico BENEVENTO; Giulia BERNARDINI; Giulia BERTAGNA; Luca BERTOGLIO; Claudio BIANCHINI MASSONI; Daniele BISSACCO; Michelangelo BISCHETTI; Luca BOCCALON; Paolo BONANNO; Stefano BONARDELLI; Raul BORIONI; Maria P. BORRELLI; Antonio BOZZANI; Umberto M. BRACALE; Stefano CAMPARINI; Aldo CANCIGLIA; Francesco CANOVA; Laura CAPOCCIA; Alessandro CAPPELLI; Pierlugi CAPPIELLO; Chiara CARLUCCIO; Alfonso CASALINO; Francesco CASELLA; Giulia CASILLI; Claudio CASTAGNO; Patrizio CASTELLI; Tommaso CASTRUCCI; Matteo CAVALLO; Carlo CAVAZZINI; Gianluca CECCANEI; Pietro CEFALI; Gianni CELORIA; Mauro CEVOLANI; Roberto CHIAPPA; Emiliano CHISCI; Carlo COMANDE; Rita COMPAGNA; Andrea CUMINO; Simone CUOZZO; Patrizia DALLA CANEVA; Ilenia D'ALESSIO; Giuseppe D'ARRIGO; Giovanni DE CARIDI;

Conflicts of interest

The authors certify that there is no conflict of interest with any financial organization regarding the material discussed in the manuscript.

All authors read and approved the final version of the manuscript.

ITALIAN DELPHI CONSENSUS ON EVAR FEASIBILITY

Gianmarco DE DONATO; Gabriele DE DONNO; Claudio DESANTIS; Francesco DE SANTIS; Alessandro DE TROIA; Tommaso DEZI; Domenico A. DIACO; Rossella DI DOMENICO; Michele DI FILIPPO; Alessia DI GIROLAMO; Carlo P. DIONISI; Ettore DINOTO; Francesco DI STEFANO; Lucia DI STEFANO; Mario D'ORIA; Andrea ESPOSITO; Davide ESPOSITO; Ludovica ETTORE; Gian F. FADDA; Gianluca FAGGIOLI; Aaron T. FARGION; Stefano FAZZINI; Nicoletta FERMANI; Giulia FERRANTE; Mauro FERRARI; Stafanio FERRARO; Ciro FERRER; Luca FERRETTO; Ilaria FICARELLI; Federico FILIPPI; Gianluigi FINO; Enzo FORLITI; Martina FORMICONI; Loris FLORA; Mauro FRESILLI; Paolo FRIGATTI; Dalmazio FRIGERIO; Alberto FROIO; Antonio FREYRIE; Sergio FURGIUELE; Roberto GABRIELLI; Andrea GAGGIANO; Luca GALASSI; Giuseppe GALZERANO; Mauro GARGIULO; Luca GARRIBOLI; Genadi G. GENADIEV; Lucia GENTILE; Alessia GIAQUINTA; Lorenzo GIBELLO; Raffaele GRANDE; Viviana GRASSI; Arnaldo IPPOLITI; Sandro IRSARA; Andrea KAHLBERG; Nikolakopoulos KONSTANTINOS; Francesco LA CORTE; Gaetano LANZA; Antonio LAURICELLA; Elisa LAZZERI; Massimo LENTI; Marco LEOPARDI; Sandro LEPIDI; Andrea LI DESTRI; Federica LOCATELLI; Chiara LOMAZZI; Francesco LOMBARDI; Antonio LORIDO; Claudia MAGGIORE; Wassim MANSOUR; Vittorio MARCLUCCI; Daniele MASCIA; Mafalda MASSARA; Giovanni MASTRANGELO; Costanza MARGHERITINI; Gabriele MASCIA; Mafalda MASSARA; Giovanni MASTRANGELO; Costanza MARGHERITINI; Gabriele MARTIATI; Eugenio MARTELLI; Ombretta MARTINELLI; Antonio MARZANO; Francesca MAURI; Daniela MAZZACARRO; Andrea MELLONI; Roberto MEZZETTI; Stefano MICHELGAGNOLI; Bruno MIGLIARA; Mattia MIGLIARI; Massimiliano MILLARELLI; Maria MISURACA; Pietro MODUGNO; Diego MONIACI; Nunzio MONTELIONE; Andrea MONTI; Nicola MONZIO-COMPAGNONI; Mario MORO; Lorenzo MORTOLA; Gaddiel MOZZETTA; Aldo MUSILLI; Giovanni NANO; Mariateresa OCCHIUTO; Fabio M. ODDI; Bernardo ORELLANA; Paola ORLANDO; Claudia PANZANO; Chiara PANZENA; Francesco PASCUCCI; Rocco PASQU4; Leonardo PASQUETTI; Edoardo PASQUI; Alberto PACAGROSSO; Claudia PANZANO; Chara PANZERA; Francesco

Article first published online: February 6, 2024. - Manuscript accepted: December 6, 2023. - Manuscript received: October 23, 2023.

Supplementary data

For supplementary materials, please see the HTML version of this article at www.minervamedica.it